Author's response to reviews

Title: The Graduate Entry Generation: a qualitative study exploring the factors influencing the career expectations and aspirations of a graduating cohort of graduate entry dental students in one London institution

Authors:

Paul Newton (paul.4.newton@kcl.ac.uk)
Lyndon Cabot (lyndon.cabot@kcl.ac.uk)
Nairn HF Wilson (nairn.wilson@kcl.ac.uk)
Jennifer E Gallagher (jenny.gallagher@kcl.ac.uk)

Version: 3 Date: 13 August 2011

Author's response to reviews:

August 2011

Dear Sir/Madam

I confirm that we have undertaken the following actions as requested and apologise that due to the holiday period there has been a delay in resubmitting our paper for publication.

1. Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.
   - The author’s positions in the institute are declared.

2. Please include an Authors’ contributions section before the Acknowledgements and Reference list.
   - The author’s contributions to the paper are outlined in the relevant section which has been added.

3. We strongly encourage you to include an Acknowledgements section between the Authors? contributions section and Reference list.
   - The author’s acknowledgments are outlined in the relevant section which has been added.

4. Reviewers comments
   - None required action

5. Formatting
   We would also request that you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at:
http://www.biomedcentral.com/info/ifora/medicine_journals
- This was undertaken and we trust that we have made all necessary changes.

And finally, on behalf of my co-authors we would like to thank the reviewers for their helpful comments on earlier drafts of this paper and for your willingness to publish our research in BMC Oral Health.

Yours faithfully

Jennifer E Gallagher